Author's response to reviews

Title: Osseointegration of zirconia implants: an SEM observation of the bone-implant interface

Authors:

Rita Depprich (depprich@med.uni-duesseldorf.de)
Holger Zipprich (depprich@med.uni-duesseldorf.de)
Michelle Ommerborn (ommerborn@med.uni-duesseldorf.de)
Eduardo Mahn (depprich@med.uni-duesseldorf.de)
Lydia Lammers (depprich@med.uni-duesseldorf.de)
Jörg Handschel (depprich@med.uni-duesseldorf.de)
Christian Naujoks (depprich@med.uni-duesseldorf.de)
Hans-Peter Wiesmann (depprich@med.uni-duesseldorf.de)
Norbert R. Kübler (depprich@med.uni-duesseldorf.de)
Ulrich Meyer (depprich@med.uni-duesseldorf.de)

Version: 4 Date: 21 October 2008

Author's response to reviews:

Dear editor,

We submit the revised manuscript “Osseointegration of zirconia implants: an SEM observation of the bone-implant interface” for publication in Head & Face Medicine.

In detail, we modified the following issues:
hypothesized (page 3, line 21)
utilizing (page 3, line 22)
sties (page 3, line 18)

As we incorporated all remarks made by the reviewers we think the revised manuscript is now ready for publication in Head & Face Medicine.

Kind regards,
The authors